



# Article Structuring Servitization-Related Capabilities: A Data-Driven Analysis

Ornella Benedettini



Citation: Benedettini, O. Structuring Servitization-Related Capabilities: A Data-Driven Analysis. *Sustainability* 2022, 14, 5478. https://doi.org/ 10.3390/su14095478

Academic Editor: Yoshiki Shimomura

Received: 29 March 2022 Accepted: 28 April 2022 Published: 3 May 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Department of Mechanics, Mathematics and Management, Polytechnic University of Bari, 70125 Bari, Italy; ornella.benedettini@poliba.it

Abstract: The existing literature investigates organizational capabilities for servitization in exploratory studies based predominantly on interviews with managers. This has led to classification frameworks that tend to reflect managerial perceptions regarding key capabilities (rather than actual firm capabilities) and in which associations and connections between capabilities remain anecdotal. This study examines the servitization-related capabilities existing in manufacturing firms that have taken strategic service initiatives and adopts a data-driven approach to explore their underlying structure. A quantitative study based on secondary data from annual report narratives is used to assess the servitization-related capabilities of 79 firms from the aerospace and defense sector and to identify the underlying factors through exploratory factor analysis. The study identifies seventeen capabilities structured into five factors: (i) management of production/delivery operations; (ii) development of valuable and sustainable offerings; (iii) identification of incentives; (iv) planning for uncertainty and change; and (v) relationship management. The study provides evidence of servitization-related capabilities in practice. By examining gaps between existing (current) capabilities and the capabilities identified in our five-factor model, business managers of aerospace and defense firms can assess the status of servitization-related capabilities at their firms and set objectives to develop such capabilities further. The study contributes to the systematic development of a reasonable and parsimonious representation of organizational capabilities for servitization, which is statistically supported and validated through empirical data.

Keywords: servitization; capabilities; framework; structure

# 1. Introduction

As product technologies continue to grow in sophistication and complexity, it will be almost impossible to maintain a competitive edge through product differentiation alone. For this reason, many manufacturing firms have shifted their business to services in addition to products—a trend often referred to as servitization or service transition [1–3]. The servitization approach involves manufacturers exploring service opportunities in the customer activity chain, expanding the boundaries of their activities and taking over service activities previously performed by the customers and/or third parties, such as product maintenance, operation, spare part management, process optimisation, certification, system integration, and financing. Based on recent surveys, 95% of B2B manufacturing companies plan to increase their service revenues [4] and 75% of those based in Europe expect delivering services to become a significantly bigger part of their business over the next few years [5].

However, despite servitization is regarded as playing a key role in the competitive stance and survival of manufacturing firms, individual firms may fail to take advantage of such opportunities due to insufficient organizational capabilities (e.g., [6]). Capabilities are widely interpreted as the foundation for the achievement of business success through the creation of value from the deployment of available resources [7]. They reflect a firm's ability to coordinate activities and make use of its resources to achieve desired goals. Implicit

in this interpretation is a recognition that, in order succeed in their service endeavors, manufacturing firms need to have the right capabilities [8]. The existence of specific organizational capabilities, or their development, is what allows firms to perform and manage the organizational processes necessary for their service business in ways that create customer value and confer competitive advantage [9].

The need to understand firm capabilities has been increasingly evident in the literature on servitization, becoming more explicit in the recent works of Raddats et al. [10] and Valtakoski and Witell [11]. A number of exploratory studies have discussed the capabilities required by manufacturing firms to successfully provide services, with several studies developing theoretical frameworks of servitization-related capabilities. With very few exceptions, most of these studies are exploratory in nature [11,12] and use case studies and interviews with managers to identify categories and typologies of organizational capabilities that are relevant to developing, selling and delivering services. Hence, while they represent managerial perceptions about different factors that influence service activities [10], prior works do not capture the capabilities existing within firms [13]. Moreover, the suggested theoretical frameworks tend to include a large number of capability items which are defined in isolation from one another [14], leading to the question about associations and connections among the identified capabilities. In view of these research gaps, the present study was set out to:

- Provide insights into the servitization-related capabilities existing within manufacturing firms that have taken strategic service initiatives;
- Contribute to a better understanding of the underlying structure of such capabilities.

These issues are investigated in a quantitative approach that aims to contribute to the systematic development of a parsimonious representation of servitization capabilities. To this end, the study adopts an existing framework which identifies a fairly comprehensive set of capabilities supposedly relevant to the provision of services by manufacturing firms. A subsequent quantitative study based on secondary data from annual report narratives is used to assess the presence of these capabilities within 79 publicly listed servitized firms from the aerospace and defence sector and to identify the underlying factors through exploratory factor analysis. This study thus addresses the call for deeper sample-based, empirical research into servitization-related capabilities (e.g., [11]).

The remainder of the paper is structured as follows. Section 2 reviews key literature on servitization-relevant capabilities and introduces the study's focus on activity-related capabilities. Section 3 outlines the data collection method, together with the theoretical framework adopted for the empirical study. Section 4 details the methodological approach taken to analyze the data and presents the results. Lastly, Section 5 discusses theoretical and managerial contributions of the study, limitations and potential directions for future research.

## 2. Background Literature

As discussed, the idea of capabilities has often been used as theoretical lens to explore how manufacturers achieve the transition to services. Drawing on established notions from the resource-based view, several studies have sought to identify sets of resources and capabilities that should underpin service endeavors. The underlying assumption is that capabilities determine a firm's level of efficiency and effectiveness in providing services [15]; that is, the more of them a firm possesses and is able to deploy, the greater the chances for successful service activities [16].

Early studies that drew attention to servitization-related capabilities focused on manufacturing firms wishing to become solution providers, offering integrated combinations of products and services as 'hybrid offerings' [17] that are tailored to address customers' specific needs. Davies [18] singled out four types of capabilities required to compete in the provision of solutions: system integration, operational services, business consulting, and financing. Few years later, Ceci and Prencipe [19] explored firm-related and environmentrelated contingency factors that may affect the relative importance of these capabilities. Similarly, Ceci and Masini [20] addressed the fit between the aforementioned capabilities and three types of heterogeneity in the firm's operating environment. These early studies, then, referred to capabilities as well-specified destinations to be achieved in creating servitized offerings. Capabilities would concern the firms' abilities to deliver the constituting elements of a solution package (i.e., finished product, operation, consulting and financing), and developing novel combinations of such capabilities would create a differentiation advantage [18]. Subsequent research delineated the particular 'productive activities' [9] (p. 61) involved in the deployment of hybrid offerings, conceptualizing service capabilities as skill elements that help manufacturers perform those activities competitively. Storbacka [21] drew on a four-step solution process (develop—create demand—sell—deliver) to categorize relevant capabilities in the solution business. Huikkola and Kohtamäki [22] identified seven key capabilities that occur in different phases of solution development and deployment, whilst Ulaga and Reinartz [17] discussed critical capabilities along the categories of: data processing and interpretation, risk assessment and mitigation, design-to-service, hybrid offering sales, and hybrid offering deployment.

The role of capabilities has been increasingly emphasized also by the general literature on servitization, where existing studies recognize numerous important types of capabilities for successful service offering, such as partnering capabilities [23], network or ecosystem management capabilities [8,11,14,24,25], service development and innovation capabilities [9,13,16,25-28], service customization capabilities [11,14], risk assessment and mitigation capabilities [8,25], customer intimacy capabilities [15,25], culture change capabilities [25], business model design capabilities [24] and capabilities in digital technologies [14,29–32]. Within this literature, several categorization frameworks have been proposed, although most of them focus on only some particular capability blocks. Not surprisingly, given the general concern with ensuring that manufacturers achieve differentiation advantages through services, the predominant focus has been on service innovation capabilities that support the introduction and viability of new services. For example, Kindström et al. [26] suggests a framework of key micro-foundations for the successful realignment of the dynamic capabilities of sensing, seizing, and reconfiguring so as to achieve a better fit with service innovation activities. Kindström and Kowalkowski [16] use a business model framework to examine the unique capabilities that product-based firms should develop and deploy to pursue service innovation. Parida et al. [27] focus on the capabilities that allow manufacturers to develop effective service innovations for global markets, and Lütjen et al. [13] develop a framework of the capabilities that allow firms to leverage their ecosystem for service innovation.

Other research dwells upon the alignment between capabilities and elements of service offerings (that is, the question of what organizational capabilities are required for manufacturers that venture into specific types of services). However, whereas some early studies have sought out to explore the whole spectrum of service elements that manufacturers may include in their offers (e.g., [17,33]), the accounts of the more recent literature have mainly centered on the capability requirement for advanced services, which address complex customer needs and are focused on delivering specified outcomes [34]. For example, Story et al. [25] adopts a multi-actor perspective and identifies categories of complementary capabilities that manufacturers, customers and intermediaries should develop to support the implementation of advanced services. Similarly, in one of the very few quantitative studies in the area, Sjödin el al. [14] use fuzzy-set comparative analysis (fsQCA) to investigate the complementary, enhancing and suppressing effects of four capabilities leading to advanced services. Still more recent works have considered the capabilities that manufacturers should possess as providers of smart services. Hasselblatt et al. [35] identifies five strategic capabilities required for a manufacturer to operate as a provider of IoT solutions. Münch et al. [32] similarly investigate the capabilities required to offer smart product-service systems and adopts socio-technical systems theory to organize them in a structured framework. Likewise, Huikkola et al. [36] examines the dynamic capabilities that facilitate strategic change from selling products to providing smart solutions.

Other recent works have considered the process of capability development. Importantly, these studies have suggested that several dynamic and operational capabilities for servitization can be developed interactively with suppliers and customers [8] and have indicated that a sequential approach to capability development (i.e., first building front-office instead of back-office service capabilities is potentially best suited for both SMEs [11] and global manufacturers) [12].

Undoubtedly, the extant literature proposes a multitude of servitization-related capabilities. Besides the explorative approach, one possible reason for this may be also that research so far has considered servitization capabilities across multiple 'hierarchical levels' [15]. Referring to Grant [37], organizational capabilities can be arranged into a hierarchy based on their level of specialization. This hierarchy spans across various organizational functions and includes several levels, starting with task-specific capabilities (at the bottom of the hierarchy) and continuing upwards with specialized, activity-related, functional and cross-functional capabilities. Lower level capabilities are progressively aggregated and integrated into higher level capabilities. While prior research on servitization tends to discuss capabilities across multiple hierarchical levels [15] (p. 158), we chose to focus exclusively on activity-related capabilities, which are mid-level in the hierarchy. By doing so, we ensure that the studied capabilities, of which we sought to establish associations and connections, do not overlap with each other. Unlike task-specific and specialized capabilities, activity-related capabilities are not specific to particular end products or services. Accordingly, we regard servitization capabilities as abilities that are important to service provision but are not necessarily exclusive to it; that is, they may, at least in principle, apply as well to product activities.

## 3. Research Method

#### 3.1. Theoretical Framework

The measurement items used in the empirical study are reported in Table 1. They were drawn from an existing framework, which was developed in the context of a multi-year study (by combining insights from the literature with interviews with senior managers at twelve leading servitized manufacturing firms) and adopted in previous research [38]. We selected this framework since it offers a comprehensive list of pertinent capabilities, grounded in research results and in the literature. Nevertheless, the framework could be adapted to develop a list of measurement items aligned with our research goals.

Table 1. Operationalization of servitization capabilities (adapted from Benedettini et al. [38]).

Category Item Description						
ECOSYSTEM AWARENESS—EA						
How well does the company know the mem	bers of its ecosystem?					
Customer perspective	Understanding of current/potential customers and their business models—EA1					
Partner perspective	Understanding of current/potential partners and the role that they play in the ecosystem—EA2					
Influencer perspective	Understanding of groups and institution that influence customers, partners, suppliers and competitors—EA3					
How well does the company understand the economics of its ecosystem?						
Value creation perspective	Understanding of who creates value in the ecosystem (and how this is likely to evolve over time)—EA4					
Value capture perspective	Understanding of who captures value in the ecosystem (and how this is likely to evolve over time)—EA5					
Power perspective	Understanding of where power lies in the ecosystem and what this implies for the ability to capture value—EA6					

Category	Item Description					
How well does the company understand th	e dynamics of its ecosystem?					
Dynamics perspective	Understanding of how the ecosystem is evolving (and who has the potential/interest to					
Skills and assets perspective	erspective Understanding of which skills and assets are in short supply in the ecosystem (and w controls access to these skills and assets)—EA8					
Competition perspective	Understanding of where competition is most intense in the ecosystem—EA9					
	VALUE PROPOSITION—VP					
How well does the company understand its	client's business model and the broader ecosystem?					
Value creation perspective	Understanding of how customers (and other significant ecosystem organizations) create value—VP1					
Value capture perspective	Understanding of how customers (and other significant ecosystem organizations) capture value—VP2					
Constraints perspective	Understanding of the constraints that customers (and other significant ecosystem organizations) face as they seek to create and capture value—VP3					
How clearly can the company articulate its	value proposition and the associated benefits?					
Customer recognition perspective	Clearly defined value proposition that customers (and other significant ecosystem organizations) understand—VP4					
Internal recognition perspective	Clearly defined value proposition that is accepted and embraced within the organization—VP5					
Cost perspective	Value proposition that is demonstrated to provide a cost-effective solution to customer problems—VP6					
Has the company clearly and unambiguously demonstrated its delivery skills in relation to the value proposition?						
Customer confidence perspective	Ability to deliver the value proposition recognized by customers (and other significant ecosystem organizations)—VP7					
Demonstrated capability perspective	Use of contracts allowing to demonstrate the ability to deliver even richer value propositions—VP8					
Pilot capability perspective	Pilot projects allowing to demonstrate the ability to deliver even richer value propositions—VP9					
	VALUE DELIVERY—VD					
How well has the company defined its valu	e proposition and designed the value delivery system?					
Internal capability perspective	Understanding of the internal capabilities required to deliver the value proposition (and how they are likely to evolve over time)—VD1					
Ecosystem capability perspective	Understanding of the capabilities needed by ecosystem partners to support the delivery of the value proposition—VD2					
Technology perspective	Understanding of the technologies required to deliver the value proposition (and how they are likely to evolve over time)—VD3					
How well has the company identified partn	ers and developed appropriate governance mechanisms?					
Partnership perspective	Ability of ecosystem partners to support the delivery and enhancement of the value proposition—VD4					
Trust perspective	Trusted relationships with ecosystem partners involved in the delivery of the value proposition—VD5					
Governance perspective	Governance mechanisms in place that encourage cooperation among the ecosystem partners involved in the delivery of the value proposition—VD6					
How well does the company coordinate mu	lti-party delivery?					
Incentive perspective	Internal incentives in place that encourage cooperation among those involved in the delivery of the value proposition—VD7					
Partnership perspective	Fair and clear dealings with ecosystem partners involved in the delivery of the value proposition—VD8					
Cultural perspective	Shared culture within the organization designed to support the delivery of the value proposition—VD9					

Category	Item Description					
ACCOUNTABILITY SPREAD—AS						
How well does the company understand the risks a	ssociated with its value delivery system?					
Performance risk perspective	Understanding of the overall performance risk inherent in the value delivery system—AS1					
Financial risk perspective	Understanding of the overall financial risk inherent in the value delivery system—AS2					
Long-term risk perspective	Understanding of the dynamic, long-term risk inherent in the value delivery system—AS3					
How good are the company's systems for measuring	g and quantifying risks?					
Measurement perspective	Use of measures for quantifying risk in the value delivery system—AS4					
Data access perspective	Access to the data needed to measure risk in the value delivery system—AS5					
Data quality perspective	Confidence in the quality of data used to measure risk in the service delivery system—AS6					
How well does the company price and flow risk to	its ecosystem partners?					
Risk ownership perspective	Understanding of who is the best owner of risk in the value delivery system—AS7					
Risk pricing perspective	Use of methods for articulating and pricing the risk inherent in the value delivery system—AS8					
Risk mitigation perspective	Understanding of how the risk inherent in the value delivery system can be mitigated—AS9					

# Table 1. Cont.

This adaptation was necessary due to the fact that, as outlined above, our study focuses on activity-related capabilities which are relevant to the service business but may, at least in principle, apply also to manufacturers' product business. More specifically, we did non focus on specialized and task-specific capabilities that deal exclusively with service provision, but on capturing more generic capabilities and skills that firms may also be able to particularize and adapt to the provision of physical products. Therefore, for example, a capability item that in the original framework reads as 'We have a deep understanding of the technologies required to deliver the service value proposition and how these technologies will evolve over time' was changed as 'Understanding of the technologies required to deliver the value proposition (and how they are likely to evolve over time)'. That is, we adapted the original capability item assuming that firms do not need differentiated activity-related abilities in order to identify, implement and exploit the technological options that are available to support the delivery of products and services. If a firm is able to understand and align with technological developments, such ability will likely apply to both product and service provision. As this example also illustrates, the adaptation of the original framework involved the rewording of certain capability items in order to simplify interpretation.

At the highest level, the framework consists of four broad categories of capabilities. It suggests that, when adopting a servitized business strategy, manufacturing firms need to innovate their *value proposition*, creating new or extended value for their customers. They do so in the context of a broader *ecosystem*, that they need to understand and leverage. To deliver the value proposition to the customers, they also often need to engage ecosystem partners, creating networks of firms with shared or pooled resources; therefore, they find it necessary to innovate their *value delivery* system. Importantly, if there is innovation in both the value proposition (offering new or extended customer value) and in the value delivery system (engage ecosystem partners in delivering this value), manufacturers are likely to increase their risk or *accountability spread*; that is, they take on more responsibility for customer outcomes, while at the same time reducing their direct control over the resources needed to deliver these outcomes. These four broad categories of capabilities—value proposition, ecosystem awareness, value delivery and accountability spread—are further detailed into twelve bundles of capabilities and 36 individual capabilities. For example,

the ecosystem awareness category includes three bundles of capabilities, tapping into the firm's knowledge of the members of the ecosystem, the firm's understanding of the economics of the ecosystem, and the firm's understanding of the dynamics of the ecosystem (cf. Table 1). Each bundle is composed of three capability items, for a total of twelve individual capabilities making up the ecosystem awareness category. For example, the firm's knowledge of the members of the ecosystem consists of capability items reflecting the firm's knowledge of customers, of partners, and of other influential actors in the ecosystem.

#### 3.2. Data Collection

## 3.2.1. Sample Selection

The empirical context for this study is the global aerospace and defense industry. Employing a single industry context minimizes potential confounds across multiple industries (e.g., different ecosystem structures, varying competitive pressures) [39,40]. The aerospace and defense sector represents a leading exemplar of the trend towards servitization, as the complexity, high unit cost, and long life of the manufactured equipment can be exploited to drive robust and smooth revenue streams from advanced forms of customer support offerings [41,42]. Therefore, the aerospace and defense industry provides a research setting in which the development of servitization-related capabilities is likely to be important. The primary industry of operation was used to search in Capital IQ for public firms from the aerospace and defense sector. The criterion of larger than 100 employees in size was used to ensure that the firms had strategized service activities. The resulting sample included 138 US and non-US firms.

#### 3.2.2. Data Collection Methodology

Information was sought about the framework capabilities by content analyzing the firms' 10-K or annual report narratives. The 10-K filing or annual report provides a comprehensive overview of a firm's undertakings, achievements and performance during the fiscal year and is a primary source of information for shareholders, investors and other observers. Investors and financial analysts, in particular, study and analyze 10-Ks and annual reports to understand the business strategies, resources and future prospects of the firm. Annual reports are considered an objective, accurate and reliable source of information [43]. Thus, any reference to service-related capabilities in the 10-K or annual report is likely to be important to the firm [41]. However, 10-K and annual report narratives are often unstructured, and extracting information from them is time-consuming and difficult. Therefore, a systematic content analysis was adopted to identify servitization-related capabilities in an accurate and effective way.

This technique helps researchers convert empirical content into theoretical concepts [22] and make 'inferences by objectively and systematically identifying specified characteristics of message' [44,45]. As a methodological approach, it gained legitimacy in management research during the 1980s for systematic evaluation of the information contained in corporate documents [46] and for drawing inferences from the textual communications of managers [47]. Although rarely used in service research, the content analysis method has become increasingly popular in various fields of supply chain and logistics research. Other business disciplines, including strategy, organizational behavior and marketing, have also leveraged content analysis tools and techniques to examine various research topics [48], such as the modification of marketing and innovation activities in response to past stock returns and volatility (e.g., [49]), discursive legitimation in merger processes (e.g., [50]) and management of media reaction after firms' wrongdoings (e.g., [51]). Given its ability to uncover evidence embedded in textual documents [45,48], content analysis was a logical choice for converting annual report narratives into the data needed for the present study.

## 3.2.3. Data Collection Execution

Although multiple sources were consulted (Capital IQ database, SEC website, company websites, email/telephone contact), the annual report was not available for all companies. Some companies simply didn't have an English version of their annual report, or only reported financial data. This reduced the sample size from 138 to 94 companies.

Capital IQ long business descriptions were then examined to understand if the companies were servitized. Benedettini and Neely [41] identified 15 categories of services that aerospace and defense companies may offer. The companies were classified as servitized if their business description provided explicit evidence that they offered one or more of these service categories to end customers. For some companies, for which Capital IQ did not provide the long business description, the information was gathered from corporate websites. At this stage, it became evident that the sample included eight companies that offered no services, (i.e., eight non-servitized manufacturers). These were removed from the sample. Nevertheless, seven companies were identified as pure service providers (i.e., companies with no manufacturing operations) and therefore also eliminated.

The content analysis was conducted with the coding software Wordstat 7 (Provalis Research, Montreal, Canada). When supported by a software tool, the content analysis methodology entails developing and applying a 'concept dictionary' of search terms, on the basis of which words are extracted from the text and presented to the analyst [52]. A wide range of search terms was selected to enable the location of the servitization-related capabilities in the companies' annual reports. These included identifiable synonyms and alternative terminology/spelling for each search term. Appropriate 'rules' were defined based on closeness of key terms within documents so as to get more accurate hits. The complete concept dictionary included 994 specifications of search terms, organized into 119 rules. The content analytic software identified the occurrences of the search terms in the annual reports. These were manually reviewed to confirm the contextual meaning, removing all irrelevant uses and occurrences relating to forward looking statements, risk factors, duplicates and negative connotations. Examples of hits that were retained after this screening are:

"We anticipate increasing competition in our core markets as a result of continued defense industry consolidation, including cross-border consolidation of competition, which has enabled companies to enhance their competitive position and ability to compete against us".

(Safran SA) for capability EA7: Understanding of how the ecosystem is evolving (and who has the potential/interest to influence the direction of evolution);

"Taranis was designed to demonstrate the Group's ability to create a system capable of undertaking sustained surveillance, marking targets, gathering intelligence and carrying out strikes in hostile territory".

(BAE Systems plc) for capability VP9: Pilot projects allowing to demonstrate the ability to deliver even richer value propositions;

"Our success also depends on our ability to provide the people, technologies, facilities, equipment and financial capacity needed to deliver those products and services with maximum efficiency".

(Northrop Grumman Corporation) for capability VD1: Understanding of the internal capabilities required to deliver the value proposition (and how they are likely to evolve over time).

"We believe that we have adopted appropriate measures to mitigate potential risks to our technology and our operations from these information technology-related and other potential disruptions".

(United Technologies Corporation) for capability AS9: Understanding of how the risk inherent in the value delivery system can be mitigated.

While qualitative content analysis would examine reference vs. non-reference to each servitization-related capability, quantitative content analysis requires to consider the frequency of reference to each capability; i.e., the number of times the capability is referred to in the text [53–55]. In line with the quantitative approach, the content analysis identified the frequency of reference to each servitization-relevant capability. Previous applications of quantitative content analysis on annual report narratives have deemed the frequency of reference to a specific concept to be a true indicator of the intensity of that concept at the firm (e.g., [56]). Therefore, we assumed that the number of mentions of each servitization-related capability indicates the level of that capability possessed by the firm.

#### 4. Results

#### 4.1. Exploratory Factor Analysis (EFA)

The analysis began with the original 36 items. Three items (EA6, VD8 and AS6) were excluded since the content analysis found no reference to them in the annual reports of the sample companies. Two items (VD2 and AS5) were further excluded due to there being no correlations with other items in excess of 10.31 [57,58]. An exploratory factor analysis was performed on the remaining items. Principal axis was employed as an extraction method to allow for departure from multivariate normality in the data [59,60]. The Kaiser-Meyer-Olkin (KMO) measure of sampling adequacy did not achieve the recommended threshold of 0.5 for all items [61]. Thus, the items with the smallest MSA values were deleted one at a time and the factor analysis was rerun until all items achieved MSA values above the threshold of 0.5 [57]. This led to the deletion of twelve items (VD5, VP1, VP2, EA8, VP3, EA1, AS8, EA4, EA2, AS3, EA9, EA3). The remaining set of items demonstrated acceptable KMO values for the overall test (0.579) as well as each individual item. Examination of partial correlations—equal to 0.404 and 0.340 for items VP4 and AS1, respectively, and below 0.3 for all other items—also indicated that 'true' factors existed in the data [57,58]. In addition, the result of the Bartlett's test of sphericity (Chi-Square = 412.235, p = 0.000) [62] confirmed that the use factor analysis was appropriate [57,58].

The common Latent Root Criterion (Guttman-Kaiser rule) suggested retaining four factors for rotation (eigenvalues > 1) [57,58,63,64]. However, this technique has been found not to be always reliable [60,63,65,66]. In particular, as noted by Hair et al. [57], if the number of variables is less than 20, the tendency is for this technique to extract too few factors. Examination of the scree plot [57–60] indicated the presence of five factors in the data. The five-factor solution also coincided with the minimum number of factors that achieved a cumulative percentage of total variance explained exceeding 100 per cent (due to the use of principal factor extraction, some negative eigenvalues were generated) [67]. Combining these results together [57,60,65,66] led to the decision to retain five factors for interpretation.

An oblique rotation method (oblimin) was applied to foster the interpretability of the extracted factors. Oblique rotation has been proven to outperform orthogonal rotation in avoiding that potential inter-correlation among the factors may produce a solution that distorts factor loadings away from simple structure [59,63,65]. Further to the rotation, item VP8 and item VD7 were excluded from the analysis due to loadings on all factors below the minimum requirement of 0.4 [68–71] All of the remaining items loaded onto a factor with a minimum factor loading of 0.464. Also, there were no cross-loading items that loaded at 0.4 or higher on more than one factor [59]. Communalities of all items were at acceptable levels ( $\geq$ 0.3) [58,72,73]. The final factor solution, which comprises a respectable 17 of the original 33 items, is presented in Table 2.

As shown in Table 2, the five factors demonstrated satisfactory internal consistency. Cronbach's alphas ranged from 0.628 to 0.7, exceeding the generally agreed lower limit of 0.6 for exploratory research [57,74–77]. Similarly, the factor solution resulted in adequate item-total correlations and inter-item correlations. For all items, the item-total correlation surpassed the recommended cut-off value of 0.3 [78] whereas the mean inter-item correlation was larger than the recommended minimum of 0.2 [79].

Item	Factor and Item Description	Factor Loading	Communality	Cronbach's Alpha	Item- Total Correlation	Mean Interitem Correlation
	Factor 1: MANAGEMENT OF PRODUCTION/			0.673		
VD9	Shared culture within the organization designed to support the delivery of the	0.745	0.560		0.623	0.242
AS4	value proposition Use of measures for quantifying risk in the value delivery system	0.464	0.310		0.378	0.389
AS7	Understanding of who is the best owner of risk in the value delivery system	0.537	0.450		0.307	0.437
AS9	Understanding of how the risk inherent in the value delivery system can be mitigated	0.637	0.480		0.534	0.292
	Factor 2: DEVELOPMENT OF VALUABLE AND			0.700		
VP6	Value proposition that is demonstrated to provide a cost-effective solution to	0.643	0.419		0.525	0.344
VP7	customer problems Ability to deliver the value proposition recognized by customers (and other significant	0.737	0.555		0.577	0.313
VD1	ecosystem organizations) Understanding of the internal capabilities required to deliver the value proposition (and how they are likely to evolve	0.613	0.468		0.478	0.373
AS1	Understanding of the overall performance risk inherent in the value delivery system	0.472	0.320		0.368	0.446
	Factor 3: IDENTIFICATION OF INCENTIVES			0.699		
EA5	Understanding of who captures value in the ecosystem (and how this is likely to evolve over time) Clearly defined value	0.639	0.437		0.466	0.502
VP4	proposition that customers (and other significant ecosystem organizations) understand Pilot projects allowing to	0.754	0.577		0.621	0.307
VP9	demonstrate the ability to deliver even richer value propositions	0.568	0.498		0.466	0.502
	Factor 4: PLANNING FOR UNCERTAINTY AND CHANGE Understanding of how the			0.628		
EA7	ecosystem is evolving (and who has the potential/interest to influence the direction of evolution) Understanding of the	0.488	0.460		0.429	0.370
VD3	technologies required to deliver the value proposition (and how they are likely to evolve over time)	0.644	0.442		0.531	0.243
AS2	Understanding of the overall financial risk inherent in the value delivery system	0.477	0.400		0.358	0.467

**Table 2.** Results of the exploratory factor analysis.

Item	Factor and Item Description	Factor Loading	Communality	Cronbach's Alpha	Item- Total Correlation	Mean Interitem Correlation
	Factor 5: RELATIONSHIP MANAGEMENT			0.638		
VP5	Clearly defined value proposition that is accepted and embraced within the organization	0.539	0.300		0.440	0.378
VD4	Ability of ecosystem partners to support the delivery and enhancement of the value proposition	0.596	0.377		0.473	0.337
VD6	Governance mechanisms in place that encourage cooperation among the ecosystem partners involved in the delivery of the value proposition	0.615	0.432		0.428	0.394

#### 4.2. Confirmatory Factor Analysis (CFA)

A confirmatory factor analysis was conducted to validate the factor model [60]. Although the *p*-value for the model was below the threshold of 0.05, other fit indexes suggested that the model provided an adequate fit to the data ( $\chi^2$  (d.f.) = 155.64 (109), RMSEA = 0.074, SRMR = 0.086) [57,63]. Specifically, the normed  $\chi^2$  ( $\chi^2$ /d.f.) was less than 3 [57,80,81], RMSEA was less than 0.08 [57,82–85], and SRMR was less than 0.1 [57,85]. Additionally, CFI and TLI were satisfactorily close to one (CFI = 0.820, TLI = 0.775), again suggesting an adequate model fit [58,83].

As Table 3 shows, all individual items' standardized coefficients from the measurement model were highly significant and 15 of the total 17 coefficients where greater than 0.5, in support of convergent validity of the factor structure [57,70,86]. The correlation matrix (Table 4) shows that none of the correlations between factors exceeded the 0.85 threshold, thus indicating that the factor model achieved adequate discriminant validity [63]. In addition, the square root of the average variance extracted (AVE) for each factor was larger than the correlations with other factors, confirming that the factors shared more variance with their measuring items than with other factors in the model [86–88]. Finally, the composite reliability scores of all factors (Table 3) were higher than the recommended value of 0.6, demonstrating internal consistency [57,89].

#### 4.3. The Final Factor Model

The empirical findings suggest that the profile of servitization-relevant capabilities in manufacturing companies from the aerospace and defense sector encompasses five core dimensions. The first dimension includes four capabilities, i.e., (i) development of an organizational culture supporting the delivery of the value proposition, (ii) assessment of risk inherent in the value delivery system, (iii) understanding of who is the best owner of such risk, and (iv) of how such risk can be mitigated. As this dimension covers capability items referring to the firm's ability to plan and manage operational activities, we label it as 'management of production/delivery operations'. The second dimension includes such capabilities as: (i) development of a value proposition providing a cost-effective solution to customer problems; (ii) achievement of external recognition for the ability to deliver the value proposition; (iii) understanding of internal skills and resources required to deliver the value proposition over time; and (iv) of the overall performance risk inherent in the value delivery system. As this dimension covers the content and viability of the value offering, we labelled it as 'development of valuable and sustainable offerings'. The third dimension incorporates three capabilities, referring to: (i) understanding of current/future value capture mechanisms within the company's ecosystem; (ii) clarity of the value proposition for customers and other significant ecosystem organizations; and (iii) running pilot projects

to demonstrate the company's ability to provide even richer value propositions. As these capabilities focus on outlining incentives for customers and other ecosystem organizations involved with the firm's service activities, we classified them as 'identification of incentives'. The fourth dimension incorporates (i) understanding of how the ecosystem is evolving and who may influence the direction of evolution; (ii) understanding of the technologies required to deliver the value proposition and how they are likely to evolve over time; and (iii) understanding of the overall financial risk inherent in the value delivery system. We classified these capabilities as 'planning for uncertainty and change'. Finally, the fifth capability dimension incorporates: (i) acceptance and embracing of the value proposition within the organization; (ii) ability of ecosystem partners to support the delivery and enhancement of the value proposition; and (iii) introduction of governance mechanisms to encourage cooperation among ecosystem partners involved in the delivery of the value proposition. Such capabilities focus on coordinating efforts of internal and external actors and were thus defined as 'relationship management' capabilities.

Table 3. Results of the confirmatory factor analysis.

	Factor	Item	Standardized Coefficient	Average Variance Extracted (AVE)	Composite Reliability (CR)
Factor 1	MANAGEMENT OF PRODUCTION/ DELIVERY OPERATIONS	VD9 AS4 AS7 AS9	0.735 *** 0.521 *** 0.348 ** 0.753 ***	0.589	0.689
Factor 2	DEVELOPMENT OF VALUABLE AND SUSTAINABLE OFFERINGS	VP6 VP7 VD1 AS1	0.646 *** 0.774 *** 0.619 *** 0.413 ***	0.613	0.712
Factor 3	IDENTIFICATION OF INCENTIVES	EA5 VP4 VP9	0.591 *** 0.782 *** 0.641 ***	0.671	0.679
Factor 4	PLANNING FOR UNCERTAINTY AND CHANGE	EA7 VD3 AS2	0.669 *** 0.583 *** 0.547 ***	0.599	0.642
Factor 5	RELATIONSHIP MANAGEMENT	VP5 VD4 VD6	0.571 *** 0.652 *** 0.602 ***	0.608	0.638

\*\* p < 0.05, \*\*\* p < 0.01. Fit indices:  $\chi^2$  (d.f.) = 155.64, d.f. = 109, RMSEA = 0.074, SRMR = 0.086, CFI = 0.820, TLI = 0.775

Table 4. Discriminant validity of the factor model.

	Factor 1	Factor 2	Factor 3	Factor 4	Factor 5
Factor 1	(0.767)				
Factor 2	-0.172	(0.782)			
Factor 3	-0.007	-0.086	(0.819)		
Factor 4	-0.507	0.233	0.626	(0.773)	
Factor 5	-0.205	-0.053	-0.118	0.262	(0.779)

Diagonals in parentheses are square roots of the average variance extracted from observed variables (items); Off diagonals are correlations between factors.

# 4.4. Interpretation of the Factor Model

Figure 1 illustrates the factor model.

Issues concerning the organization and management of service operations are mentioned frequently in the servitization literature [90,91] and several studies have highlighted that manufacturing firms need operations management capabilities in order to improve the quality and reduce the costs of service operations and processes (e.g., [11,17,92]). The first dimension in our model identifies relevant operations management capabilities in two areas. The first area refers to the organization's culture. Corporate culture influences the values and behaviors of employees [93]. A service-oriented culture has been frequently discussed in the literature as key to encourage service personnel to adapt service processes to the needs, problems and issues of individual customers [94]. A service-oriented culture also facilitates the strong integration among back-office and front-office service units necessary for achieving cost-efficiency without compromising the ability to offer service excellence and customization [90]. The second capability area captured by the operations management dimension of our model refers to risk management. The presence of three capability items in this area strongly reflects the aerospace and defense context of the study sample. Typically, aerospace and defense customers ask for availability-based service contracts where through-life support of high-technology, complex and long-life products/equipment is contracted out on an availability basis [42]. Such contracts involve significant operational risks that service suppliers need to take on and manage as an integral part of the offering [95]. The capability items in this area indirectly capture also the complexity of service delivery operations in the aerospace and defense sector [96], which requires servitized manufacturers to co-operate with service partners, and hence to elaborate ways to share and distribute operational risks.



Figure 1. Illustration of the factor model.

The importance for a company to devise value propositions that provide effective customer value is not a new insight. The second dimension in our model, focusing on capabilities related to the design of value offerings, confirms this importance for servitized firms. As observed by Johnstone et al. [97], aerospace and defense manufacturers especially need to understand value through the eyes of the customers so as to develop products and services as seamless, comprehensive and largely customer-specific propositions. Intriguingly, this dimension further includes three capability items that point to the firm's capacity to sustain or "afford" the delivery of the value proposition. As previously outlined, in the aerospace and defense industry, servitized manufacturers often have to take responsibility for complex service performance, as well as providing high-technology products. Achieving standardization and efficiency is particularly difficult in such circumstances, and lack of specific competences and resources may jeopardize the firm's ability to deliver the contract value proposition [97]. Nevertheless, it is worth noting that one of the three capability items refers to a perceptual dimension of value [98]; that is, the firm's ability to gain the trust of customers (and of other actors in the service network) in the viability of its value proposition. Penttinen and Palmer [99] identify this ability as playing an important role in the transition from product sales to total-service contracts such as those that are often implemented in aerospace and defense [96].

The capability items collectively constituting the third dimension of our model point to alignment of incentives. Prior research underscores that servitized manufacturers struggle with differences between service-based value creation and traditional sale of products [100]. Scholars have highlighted that, as a consequence, servitized manufacturers need to explicitly define and demonstrate how their offerings would create value for customers and other relevant actors in order to incentivize them to embrace such offerings [72]. In aerospace and defense contracting, incentives are used to ensure that the objectives of the actors involved are aligned [95]. In this respect, the capability to establish pilot projects may help develop and sustain customer trust and similar relationships with other parties. Likewise, pilot projects and clearly defined value propositions may encourage customers and other actors to experiment with new offerings and approaches [101].

The fourth dimension in our model, concerning issues of uncertainty and change, conveys the notion that servitized companies from aerospace and defense operate in complex and dynamic environments, and need explicit capabilities to deal with this. According to our model, three factors are responsible for the capability challenges in this area. The first factor concerns the high technological requirements and technological commoditization that characterize the sector [97]. Technological developments in both products and service technologies can increase the capabilities of servitized providers and, if adequately exploited, can positively influence their operations and value delivery strategies [95]. However, they also introduce risks of rapid obsolescence of products and digital services. The second factor reflects the complex and highly regulated business environment. Servitized companies need to be proactive and flexible in order to keep up with such things as changes in procurement policies of military and commercial clients, regulatory requirements, demand trends, and changes in availability of productive resources [97]. The third factor recalls the service delivery uncertainty that may be driven by service operations that are difficult to standardize and require input from multiple parties [95]. Since service performance depends on the outcome as well as the process of service delivery, understanding this uncertainty plays a key role in the formulation of the service strategy.

Finally, the fifth dimension in our model captures the well-known issue of managing internal and external relationships. This study's findings align with the literature suggesting that servitized companies need to achieve internal coordination (e.g., [16,102]), for example between the sales and field service departments. Clearly, when employees understand the value proposition as a whole and where they fit into the delivery effort, they are encouraged to think outside their functional role and increase their cooperation and interaction with other functional areas [97]. Likewise, an important managerial challenge is the coordination and prioritization of relationships with external partners contributing

the delivery of the value proposition. Poor decisions regarding partner selection have been previously identified as a potential cause of failed servitization [103–105]. Our findings confirm the relevance of a firm's ability to select partners that complement the firm's own competences in a fruitful way [106]. As one would expect given the aerospace and defense context of the study, and the associated intricate contractual settings of service provision, our model further outlines that relevant external coordination capabilities further involve the ability to establish relationship governance arrangements with business partners. Sjödin et al. [107] remark that these governance arrangements, relying on either structural or social mechanisms, define how service partners should behave in the customer relationship and are crucial to manage service processes for effectiveness, productivity and low transaction costs.

# 5. Discussion and Conclusions

#### 5.1. Theoretical Contribution

The study makes the following contributions to the literature. First, we extend the existing literature on servitization by providing empirical evidence of servitization-related capabilities in practice. With few recent exceptions using surveys [11,13,14], previous research into the capabilities that are needed by manufacturing firms to effectively manage a service business has been based on case studies and interviews with managers. Previous research, therefore, relies on managerial perceptions about the abilities and skills that should underlie service provision. Our study, instead, assesses capability constructs using secondary data from annual report narratives and hence it captures the capabilities existing within manufacturing firms that have moved into service provision. We found evidence referring to 33 of the 36 capability items in our original framework. Therefore, our results add to the findings of Valtakoski and Witell [11] by showing that servitization capabilities can be operationalized and measured also using secondary data. Second, by exploring the underlying structure of our empirical data, we contribute to prior literature in relation to structuring servitization-relevant capabilities. While the extant literature discusses servitization capabilities comprehensively and provides several classification frameworks, these are based on associations and relationships between capabilities that remain anecdotal. Such frameworks have been extremely valuable in advancing theoretical development, but they tend to be heterogeneous and complex, as they include a large number of capability items which are defined in isolation from one another [14]. A distinct contribution of our study is therefore the achievement of a parsimonious representation of servitization capabilities, which is statistically supported and validated through empirical data. In this representation, servitization capabilities in manufacturing firms are comprised of five factors: (i) management of production/delivery operations; (ii) development of valuable and sustainable offerings; (iii) identification of incentives; (iv) planning for uncertainty and change; and (v) relationship management. We contend that these factors should be considered the backbone of service capabilities for manufacturing firms from the study's industrial sector.

#### 5.2. Managerial Contribution

The results of this study also have managerial implications. The presented five-factor model and its capability items can serve as a managerial navigator to assess the current status of servitization capabilities and prioritize development activities. By examining gaps between existing (current) capabilities and the capabilities appearing in the model, business managers at aerospace and defense firms can analyze the maturity of their firms for services and, more importantly, set objectives to develop their service capabilities that managers within manufacturing firms may think to address in their strategic initiatives for service business development. Our findings indicate on which capability areas managers of servitized aerospace and defense firms should concentrate their attention and resources. In particular, managers should address the management of production/delivery

operations through developing a supportive organizational culture and the ability to assess, allocate and manage operational risks. They should also steer their strategic initiatives towards the achievement of valuable and sustainable offerings through developing an externally recognized ability to build a value proposition that addresses customer needs, in parallel with an ability to understand the internal capabilities and the risks involved in delivering such value proposition. In addition, managers should concentrate on outlining incentives for customers and other ecosystem organizations through developing the abilities to understand current/future value capture mechanisms within the ecosystem, to define a value proposition that is widely understood within the ecosystem, and to run pilot projects demonstrating the viability of even richer value propositions. Business managers should also give attention to planning for uncertainty and change through developing the abilities to understand how their ecosystem and the technologies involved in the delivery of the value proposition may evolve, and to understand the financial risks involved in the value delivery system. Likewise, managers should address the management of internal and external relationships through developing the abilities to disseminate the value proposition within the organization, to involve ecosystem partners that may support the delivery/enhancement of the value proposition, and to set-up governance mechanisms encouraging ecosystem-wide cooperation in value delivery. Second, the five-factor model can help managers of aerospace and defense firms divide servitization-related capabilities into logical groups. This assessment is useful for understanding the rationale for developing, acquiring or maintaining individual capabilities, as well as for evaluating substitutive and complementary relationships among capabilities when service business development activities are strategized.

## 5.3. Limitations and Future Research Opportunities

As with all research, this study carries some limitations. First, we examine servitizationrelevant capabilities in the context of a single industry. Care must be taken in generalizing our results to any servitized manufacturer. Our results are based on a sample of firms from the aerospace and defense industry, where the adoption of digital technologies, outcomebased contracts and multi-party service delivery is taking center stage in service provision, and which may be different from other, for example, less technological and less demanding industries. A natural progression would be a replication in other situations and/or industries yielding different specificities and offering a noteworthy contrast to our empirical context. Hence, firms from other manufacturing industries should be included in future studies to determine whether our model of servitization capabilities is replicable to other settings, or whether different capabilities emerge. Second, based on extant literature, we make the implicit assumption of a causal link between servitization-related capabilities and firm performance. However, future research could test the capabilities in our model to determine whether they actually lead to improved firm performance. Relatedly, by combining capability measures with performance measures, future research could also evaluate which of the capability groups are more important to firm performance. Finally, from a methodological perspective, we followed several previous studies (e.g., [108–110]) and performed the CFA on the same sample as the initial EFA which identified our capability factors. While the limited size of the available data did not permit to adopt a split-sample approach, we acknowledge that it is generally more sensible to use different samples for EFA and CFA.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The author declares no conflict of interest.

# References

- 1. Oliva, R.; Kallenberg, R. Managing the transition from products to services. Int. J. Serv. Ind. Manag. 2003, 14, 160–172. [CrossRef]
- 2. Raddats, C.; Kowalkowski, C.; Benedettini, O.; Burton, J.; Gebauer, H. Servitization: A contemporary thematic review of four major research streams. *Ind. Mark. Manag.* **2019**, *83*, 207–223. [CrossRef]
- 3. Vandermerwe, S.; Rada, J. Servitization of business: Adding value by adding services. Eur. Manag. J. 1988, 6, 314–324. [CrossRef]
- Simon-Kucher & Partners. Global Servitization Survey: Weather the Storm with Service-Based Strategies. Available online: https://www.simon-kucher.com/en/blog/global-servitization-survey-weather-storm-service-based-strategies (accessed on 28 March 2022).
- 5. PA Consulting. From Products to Services: Creating Sustainable Growth in Industrial Manufacturing through Servitization. Available online: http://www2.paconsulting.com/rs/526-HZE-833/images/PA%20Servitization%20Report.pdf (accessed on 28 March 2022).
- 6. Niu, Y.; Zhang, Z.; Gang, N.; Jiang, S. Disclosing the formation and value creation of servitization through influential factors: A systematic review and future research agenda. *Int. J. Prod. Res.* **2021**, *59*, 7057–7089. [CrossRef]
- 7. Day, G.S. The capabilities of market-driven organizations. J. Mark. 1994, 58, 37–52. [CrossRef]
- 8. Raddats, C.; Zolkiewski, J.; Story, V.M.; Burton, J.; Baines, T.; Ziaee Bigdeli, A. Interactively developed capabilities: Evidence from dyadic servitization relationships. *Int. J. Oper. Prod. Manag.* **2017**, *37*, 382–400. [CrossRef]
- 9. Spring, M.; Araujo, L. Beyond the service factory: Service innovation in manufacturing supply networks. *Ind. Mark. Manag.* 2013, 42, 59–70. [CrossRef]
- Raddats, C.; Burton, J.; Ashman, R. Resource configurations for services success in manufacturing companies. *J. Serv. Manag.* 2015, 26, 97–116. [CrossRef]
- 11. Valtakoski, A.; Witell, L. Service Capabilities and Servitized SME Performance: Contingency on Firm Age. *Int. J. Oper. Prod. Manag.* **2018**, *38*, 1144–1164. [CrossRef]
- 12. Jovanovic, M.; Raja, J.Z.; Visnjic, I.; Wiengarten, F. Paths to service capability development for servitization: Examining an internal service ecosystem. *J. Bus. Res.* 2019, *104*, 472–485. [CrossRef]
- 13. Lütjen, H.; Schultz, C.; Tietze, F.; Urmetzer, F. Managing ecosystems for service innovation: A dynamic capability view. *J. Bus. Res.* **2019**, *104*, 506–519. [CrossRef]
- 14. Sjödin, D.R.; Parida, V.; Kohtamäki, M. Capability configurations for advanced service offerings in manufacturing firms: Using fuzzy set qualitative comparative analysis. *J. Bus. Res.* **2016**, *69*, 5330–5335. [CrossRef]
- 15. Gebauer, H.; Saul, C.J.; Haldimann, M.; Gustafsson, A. Organizational capabilities for pay-per-use services in product-oriented companies. *Int. J. Prod. Econ.* **2017**, *192*, 157–168. [CrossRef]
- 16. Kindström, D.; Kowalkowski, C. Service innovation in product-centric firms: A multinational business model perspective. *J. Bus. Ind. Mark.* **2014**, *29*, 96–111. [CrossRef]
- 17. Ulaga, W.; Reinartz, W.J. Hybrid offerings: How manufacturing firms combine goods and services successfully. *J. Mark.* **2011**, *75*, 5–23. [CrossRef]
- 18. Davies, A. Moving base into high-value integrated solutions: A value stream approach. *Ind. Corp. Change* **2004**, *13*, 727–756. [CrossRef]
- 19. Ceci, F.; Prencipe, A. Configuring capabilities for integrated solutions: Evidence from the IT sector. *Ind. Innov.* **2008**, *15*, 277–296. [CrossRef]
- 20. Ceci, F.; Masini, A. Balancing specialized and generic capabilities in the provision of integrated solutions. *Ind. Corp. Change* **2011**, 20, 91–132. [CrossRef]
- Storbacka, K. A solution business model: Capabilities and management practices for integrated solutions. *Ind. Mark. Manag.* 2011, 40, 699–711. [CrossRef]
- 22. Huikkola, T.; Kohtamäki, M. Solution providers' strategic capabilities. J. Bus. Ind. Mark. 2017, 32, 752–770. [CrossRef]
- 23. Kohtamäki, M.; Partanen, T.; Parida, V.; Wincent, J. Non-linear relationship between industrial service offering and sales growth: The moderating role of network capabilities. *Ind. Mark. Manag.* **2013**, *42*, 1374–1385. [CrossRef]
- 24. Parida, V.; Sjödin, D.; Wincent, J.; Kohtamäki, M. Mastering the transition to product-service provision insights into business models, learning activities, and capabilities results. *Res. Technol. Manag.* **2014**, *57*, 44–53. [CrossRef]
- 25. Story, V.M.; Raddats, C.; Burton, J.; Zolkiewski, J.; Baines, T. Capabilities for advanced services: A multi-actor perspective. *Ind. Mark. Manag.* **2017**, *60*, 54–68. [CrossRef]
- 26. Kindström, D.; Kowalkowski, C.; Sandberg, E. Enabling service innovation: A dynamic capabilities approach. *J. Bus. Res.* **2013**, 66, 1063–1073. [CrossRef]
- 27. Parida, V.; Sjödin, D.R.; Lenka, S.; Wincent, J. Developing global service innovation capabilities: How global manufacturers address the challenges of market heterogeneity. *Res. Technol. Manag.* **2015**, *58*, 35–44. [CrossRef]
- 28. Wallin, J.; Parida, V.; Isaksson, O. Understanding product-service system innovation capabilities development for manufacturing companies. *J. Manuf. Technol. Manag.* 2015, 26, 763–787. [CrossRef]
- Lenka, S.; Parida, V.; Wincent, J. Digitalization capabilities as enablers of value co-creation in servitizing firms. *Psychol. Mark.* 2017, 34, 92–100. [CrossRef]
- Marcon, E.; Marcon, A.; Ayala, N.F.; Frank, A.G.; Story, V.; Burton, J.; Raddats, C.; Zolkiewski, J. Capabilities supporting digital servitization: A multi-actor perspective. *Ind. Mark. Manag.* 2022, 103, 97–116. [CrossRef]

- 31. Kamalaldin, A.; Linde, L.; Sjödin, D.; Parida, V. Transforming provider-customer relationships in digital servitization: A relational view on digitalization. *Ind. Mark. Manag.* 2020, *89*, 306–325. [CrossRef]
- 32. Münch, C.; Marx, E.; Benz, L.; Hartmann, E.; Mazner, M. Capabilities of digital servitization: Evidence from the socio-technical systems theory. *Technol. Forecast. Soc.* **2022**, *176*, 121361. [CrossRef]
- Paiola, M.; Saccani, N.; Perona, M.; Gebauer, H. Moving from products to solutions: Strategic approaches for developing capabilities. *Eur. Manag. J.* 2013, *31*, 390–409. [CrossRef]
- Baines, T.S.; Lightfoot, H.W. Made to Serve: How Manufacturers Can Compete through Servitization and Product Service Systems; John Wiley & Sons: Hoboken, NJ, USA, 2013.
- 35. Hasselblatt, M.; Huikkola, T.; Kohtamäki, M.; Nickell, D. Modelling manufacturer's capabilities for the Internet of Things. *J. Bus. Ind. Mark.* 2018, 33, 822–836. [CrossRef]
- 36. Huikkola, T.; Kohtamäki, M.; Ylimäki, J. Becoming a smart solution provider: Reconfiguring a product manufacturer's strategic capabilities and processes to facilitate business model innovation. *Technovation* **2022**, 102498. [CrossRef]
- Grant, R.M. Prospering in dynamically-competitive environments: Organizational capability as knowledge integration. *Organ. Sci.* 1996, 7, 375–387. [CrossRef]
- Benedettini, O.; Davies, J.; Neely, A. A capability-based view of service transitions. In Proceedings of the 22nd EurOMA Conference, Neuchatel, Switzerland, 26 June–1 July 2015. Available online: https://cambridgeservicealliance.eng.cam.ac.uk/ resources/Downloads/Monthly%20Papers/2015April\_Capabilitybasedviewofservicetransition.pdf (accessed on 28 March 2022).
- Menor, L.J.; Roth, A.V.; Mason, C.H. Agility in retail banking: A numerical taxonomy of strategic service groups. *Manuf. Serv.* Oper. 2001, 3, 273–292. [CrossRef]
- Zhao, X.; Yeung, A.C.L.; Lee, T.S. Quality management and organizational context in selected service industries of China. J. Oper. Manag. 2004, 22, 575–587. [CrossRef]
- 41. Benedettini, O.; Neely, A. Investigating a revised service transition concept. Serv. Bus. 2018, 12, 701–730. [CrossRef]
- Kapletia, D.; Probert, D. Migrating from products to solutions: An exploration of system support in the UK defense industry. *Ind. Mark. Manag.* 2010, 39, 582–592. [CrossRef]
- Nadkarni, S.; Narayanan, V.K. Strategic schemas, strategic flexibility, and firm performance: The moderating role of industry clockspeed. *Strateg. Manag. J.* 2007, 28, 243–270. [CrossRef]
- 44. Holsti, O.R. Content Analysis for the Social Sciences and Humanities; Addison-Wesley: Reading, MA, USA, 1969.
- 45. Krippendorf, K. Content Analysis: An Introduction to Its Methodology, 2nd ed.; Sage Publications: London, UK, 2004.
- 46. Bowman, E.H. Content analysis of annual reports for corporate strategy and risk. Interfaces 1984, 14, 61–71. [CrossRef]
- 47. D'Aveni, R.A.; MacMillan, I.C. Crisis and the content of managerial communications: A study of the focus of attention of top managers in surviving and failing firms. *Admin. Sci. Quart.* **1990**, *35*, 634–657. [CrossRef]
- Tangpong, C. Content analytic approach to measuring constructs in operations and supply chain management. *J. Oper. Manag.* 2011, 29, 627–638. [CrossRef]
- Chakravarty, A.; Grewal, R. The stock market in the driver's seat! Implications for R&D and Management. *Manag. Sci.* 2011, 57, 1594–1609. [CrossRef]
- 50. Vaara, E.; Monin, F. A recursive perspective on discursive legitimation and organizational action in mergers and acquisitions. *Organ. Sci.* **2010**, *21*, 3–22. [CrossRef]
- Zavyalova, A.; Pfarrer, M.D.; Reger, R.K.; Shapiro, L.D. Managing the message: The effects of firm actions and industry spillovers on media coverage following wrongdoing. *Acad. Manag. J.* 2012, 55, 1079–1101. [CrossRef]
- 52. Neuerdorf, K.A. The Content Analysis Guidebook; Sage Publications: Thousand Oaks, CA, USA, 2002.
- 53. Berelson, B. Content Analysis in Communication Research; Free Press: Glencoe, IL, USA, 1952.
- 54. Weber, R.P. Basic Content Analysis; Sage Publications Inc.: Newbury Park, CA, USA, 1990.
- 55. Harwood, T.G.; Garry, T. An overview of content analysis. Mark. Rev. 2003, 3, 479–498. [CrossRef]
- Davies, J.; Jogleka, N. Supply chain integration, product modularity and market valuation: Evidence from the solar energy industry. *Prod. Oper. Manag.* 2013, 22, 1494–1508. [CrossRef]
- 57. Hair, J.W.; Black, B.; Babin, R.; Anderson, R.; Tatham, R. *Multivariate Data Analysis*, 7th ed.; Pearson Prentice Hall: Upper Saddle River, NJ, USA, 2010.
- 58. Tabachnick, B.G.; Fidell, L.S. Using Multivariate Statistics, 5th ed.; Allyn & Bacon/Pearson: Boston, MA, USA, 2007.
- 59. Costello, A.B.; Osborne, J.W. Best practices in exploratory factor analysis: Four recommendations for getting the most from your analysis. *Pract. Assess. Res. Eval.* 2005, 10, 7. [CrossRef]
- 60. Fabrigar, L.R.; Wegener, D.T.; MacCallum, R.C.; Strahan, E.J. Evaluating the use of exploratory factor analysis in psychological research. *Psychol. Methods* **1999**, *4*, 272–299. [CrossRef]
- 61. Kaiser, H.F. A second generation little jiffy. Psychometrika 1970, 35, 401–416. [CrossRef]
- 62. Bartlett, M.S. Tests of significance in factor analysis. Brit. J. Psychol. 1950, 3, 77–85. [CrossRef]
- 63. Brown, T.A. Confirmatory Factor Analysis for Applied Research; The Guilford Press: New York, NY, USA, 2006.
- 64. Thompson, B. Exploratory and Confirmatory Factor Analysis; American Psychological Association: Washington, DC, USA, 2004.
- 65. Conway, J.M.; Huffcutt, A.I. A review and evaluation of exploratory factor analysis practices in organizational research. *Organ. Res. Methods* **2003**, *6*, 147–168. [CrossRef]

- 66. Ford, J.K.; MacCallum, R.C.; Tait, M. The application of exploratory factor analysis in applied psychology: A critical review and analysis. *Pers. Psychol.* **1986**, *39*, 291–314. [CrossRef]
- 67. Rencher, A.C. Methods of Multivariate Analysis, 2nd ed.; John Wiley & Sons: New York, NY, USA, 2002.
- 68. Antioco, M.; Moenaert, K.R.; Lindgreen, A.; Wetzels, M.G.M. Organizational antecedents to and consequences of service business orientation in manufacturing companies. *J. Acad. Market. Sci.* **2008**, *36*, 337–358. [CrossRef]
- 69. Hinkin, T.R. A review of scale development practices in the study of organizations. J. Manag. 1995, 21, 967–988. [CrossRef]
- Peterson, R.A. A meta-analysis of variance accounted for and factor loadings in exploratory factor analysis. *Mark. Lett.* 2000, 11, 261–275. [CrossRef]
- 71. Verma, R.; Young, S.T. Configurations of low-contact services. J. Oper. Manag. 2000, 18, 643–661. [CrossRef]
- 72. Klein, M.M.; Biehl, S.S.; Friedli, T. Barriers to smart services for manufacturing companies—An exploratory study in capital goods industry. *J. Bus. Ind. Mark.* 2018, *33*, 846–856. [CrossRef]
- 73. Partanen, J.; Kohtamäki, M.; Parida, V.; Wincent, J. Developing and validating a multi-dimensional scale for operationalizing industrial service offering. *J. Bus. Ind. Mark.* 2017, *32*, 295–309. [CrossRef]
- 74. Churchill, G.A. A paradigm for developing better measures of marketing constructs. J. Mark. Res. 1979, 16, 64–73. [CrossRef]
- John, O.P.; Benet-Martinez, V. Measurement: Reliability, construct validation, and scale construction. In *The Handbook of Research Methods in Personality and Social Psychology*; Reis, H.T., Judd, C.M., Eds.; Cambridge University Press: Cambridge, UK, 2000; pp. 339–369.
- 76. Nunnally, J. Psychometric Theory; McGraw Hill: New York, NY, USA, 1978.
- Pedhazur, E.J.; Schmelkin, L.P. Measurement, Design, and Analysis: An Integrated Approach; Lawrence Erlbaum Associates: Hillsdale, NJ, USA, 1991.
- 78. Nurosis, M. Statistical Data Analysis; SPSS Inc: Chicago, IL, USA, 1994.
- 79. Briggs, S.R.; Cheek, J.M. The role of factor analysis in the development and evaluation of personality scales. *J. Pers.* **1986**, *54*, 106–148. [CrossRef]
- 80. Bollen, K.A. Structural Equations with Latent Variables; John Wiley & Sons: New York, NY, USA, 1989.
- Marsh, H.W.; Hocevar, D. The application of confirmatory factor analysis to the study of self-concept: First- and higher-order factor models and their invariance across groups. *Psychol. Bull.* 1985, 97, 562–582. [CrossRef]
- 82. Browne, M.; Cudeck, R. Alternative ways of assessing model fit. In *Testing Structural Equation Models*; Bollen, K.A., Long, J.S., Eds.; Sage: Newbury Park, CA, USA, 1993; pp. 136–161.
- 83. Hox, J.J. Multilevel Analysis: Techniques and Application, 2nd ed.; Routledge: New York, NY, USA, 2010.
- 84. Kline, R.B. Principles and Practice of Structural Equation Modeling, 2nd ed.; Guilford Press: New York, NY, USA, 2005.
- 85. MacCallum, R.C.; Browne, M.W.; Sugawara, H.M. Power analysis and determination of sample size for covariance structure modelling. *Psychol. Methods* **1996**, *1*, 130–149. [CrossRef]
- 86. Chin, W. The partial least squares approach to structural equation modelling. In *Modern Business Research Methods*; Marcoulides, G.A., Ed.; Erlbaum: Mahwah, NJ, USA, 1998; pp. 295–336.
- Fornell, C.; Larcker, D.F. Evaluating structural equation models with unobservable variables and measurement error. *J. Mark. Res.* 1981, 18, 39–50. [CrossRef]
- Hulland, J. Use of partial least squares (PLS) in strategic management research: A review of four recent studies. *Strateg. Manag. J.* 1999, 20, 195–204. [CrossRef]
- 89. Bagozzi, R.P.; Yi, Y. On evaluation of structural equation models. J. Acad. Mark. Sci. 1988, 16, 74–94. [CrossRef]
- Hakanen, T.; Helander, N.; Valkokari, K. Servitization in global business-to-business distribution: The central activities of manufacturers. *Ind. Mark. Manag.* 2017, 63, 167–178. [CrossRef]
- 91. Rabetino, R.; Harmsen, W.; Kohtamäki, M.; Sihvonen, J. Structuring servitization-related research. Int. J. Oper. Prod. Manag. 2018, 38, 350–371. [CrossRef]
- Kowalkowski, C.; Kindström, D.; Alejandro, T.; Brege, S.; Biggemann, S. Service infusion as agile incrementalism in action. J. Bus. Res. 2012, 65, 765–772. [CrossRef]
- 93. Homburg, C.; Fassnacht, M.; Guenther, C. The role of soft factors in implementing a service-oriented strategy in industrial marketing companies. *J. Bus. Bus. Mark.* 2003, *10*, 23–51. [CrossRef]
- 94. Kohtamäki, M.; Hakala, H.; Partanen, J.; Parida, V.; Wincent, J. The performance impact of industrial services and service orientation on manufacturing companies. *J. Serv. Theory Pract.* **2015**, *25*, 463–485. [CrossRef]
- 95. Datta, P.P.; Roy, R. Operations strategy for the effective delivery of integrated industrial product-service offerings: Two exploratory defence industry case studies. *Int. J. Oper. Prod. Manag.* 2011, *31*, 579–603. [CrossRef]
- Johnstone, S.; Dainty, A.; Wilkinson, A. In search of 'product-service': Evidence from aerospace, construction, and engineering. Serv. Ind. J. 2008, 28, 861–875. [CrossRef]
- Johnstone, S.; Dainty, A.; Wilkinson, A. Integrating products and services through life: An aerospace experience. *Int. J. Oper. Prod. Manag.* 2009, 29, 520–538. [CrossRef]
- 98. Grönroos, C.; Helle, P. Adopting a service logic in manufacturing: Conceptual foundation and metrics for mutual value creation. *J. Serv. Manag.* **2010**, *21*, 564–590. [CrossRef]
- Penttinen, E.; Palmer, J. Improving firm positioning through enhanced offerings and buyer-seller relationships. *Ind. Mark. Manag.* 2007, 36, 552–564. [CrossRef]

- Brax, S.A.; Jonsson, K. Developing integrated solution offerings for remote dianostics: A comparative case study of two manufacturers. Int. J. Oper. Prod. Manag. 2009, 29, 539–560. [CrossRef]
- 101. Davies, A. Charting a path towards integrated solutions. MIT Sloan Manag. Rev. 2006, 47, 39–48.
- 102. Matthyssens, P.; Vandenbempt, K. Moving from basic offerings to value-added solutions: Strategies, barriers and alignment. *Ind. Mark. Manag.* **2008**, *37*, 316–328. [CrossRef]
- 103. Valtakoski, A. Explaining servitization failure and deservitization: A knowledge-based perspective. *Ind. Mark. Manag.* 2017, 60, 138–150. [CrossRef]
- Weigel, S.; Hadwich, K. Success factors of service networks in the context of servitization—Development and verification of an impact model. *Ind. Mark. Manag.* 2018, 74, 254–275. [CrossRef]
- 105. Bustinza, O.F.; Lafuente, E.; Rabetino, R.; Vaillant, Y.; Vendrell-Herrero, F. Make- or-buy configurational approaches in productservice ecosystems and performance. *J. Bus. Res.* 2019, 104, 393–401. [CrossRef]
- 106. Jaakkola, E.; Hakanen, T. Value co-creation in solution networks. Ind. Mark. Manag. 2013, 42, 47–58. [CrossRef]
- 107. Sjödin, D.R.; Parida, V.; Kohtamäki, M. Relational governance strategies for advanced service provision: Multiple paths to superior financial performance in servitization. *Ind. Mark. Manag.* **2019**, *43*, 113–125. [CrossRef]
- Azadegan, A.; Patel, P.C.; Zangoueinezhad, A.; Linderman, K. The effect of environmental complexity and environmental dynamism on lean practices. J. Oper. Manag. 2013, 31, 193–212. [CrossRef]
- 109. Oliva, R.; Gebauer, H.; Brann, J.M. Separate or integrate? Assessing the impact of separation between product and service business on service performance in product manufacturing firms. *J. Bus. Bus. Mark.* **2012**, *19*, 309–334. [CrossRef]
- Villena, V.H.; Revilla, E.; Choi, T.Y. The dark side of buyer–supplier relationships: A social capital perspective. *J. Oper. Manag.* 2011, 29, 561–576. [CrossRef]